

Author's response to reviews

Title: Effect of statin therapy on the progression of coronary atherosclerosis

Authors:

Jinwei Tian PhD (tianjinweidr2009@163.com)

Xia Gu MD (shijikuihua@126.com)

Yanli Sun MD (yanli2010drsun@163.com)

Xiang Ban MD (banxiang5810@sina.com)

Yun Xiao PhD (yunxiao@163.com)

Sining Hu MD (sininghudr@163.com)

Bo Yu PhD (yubodr2009@163.com)

Version: 4 **Date:** 14 April 2012

Author's response to reviews:

April 14, 2012

Editor

BMC Cardiovascular Disorders

Manuscript ID: 9159579096525455

Manuscript TITLE: Effect of statin therapy on the progression of coronary atherosclerosis

Dear editor:

Thank you very much for the review and constructive suggestions. We have tried to answer point-by-point the comments and questions raised by the reviewer. The changes were incorporated in the revised manuscript.

We believe that our manuscript has improved after the revision and hope that it now meets the expectation of the editor and the reviewers.

If you have any questions, please do not hesitate to contact me.

We added Yanli Sun and Xiang Ban as co-authors for their sufficient contribution in the revised manuscript.

Hope the answers provide enough details to fulfill expectations of the Editorial Board.

Looking forward to a positive decision,

Sincerely yours,

Bo Yu, MD. PhD